Author's response to reviews

Title: Is the state of health of rheumatoid arthritis patients receiving adequate treatment, predictable? - Results of a survey

Authors:

Rudolf Puchner (rudolf.puchner@cc-net.at)
Hans-Peter Brezinschek (hans-peter.brezinsek@meduni-graz.at)
Josef Fritz (josef.fritz@i-med.ac.at)
Manfred Herold (manfred.herold@i-med.ac.at)
Monika Mustak (monika.mustak@wienkav.at)
Thomas Nothnagl (thomas.nothnagl@stockerau.lknoe.at)
Stephan E Puchner (stephan.puchner@meduniwien.ac.at)
Andrea Studnick-Benke (a.studnicka-benke@salk.at)
Burkhard F Leeb (leeb.rheuma@aon.at)

Version: 6 Date: 26 April 2015

Author's response to reviews:

Burkhard Leeb, MD, PhD

Dear Mrs. de Guzman,

thanks for your mail. We've made the changes requested and hope that the MS can now found acceptable for publication in BMC Musculoskeletal Disorders.

Kind regards,

Burkhard Leeb, Md, PhD